Author's response to reviews

Title: Prevalence and determinants of the use of self-tests by members of the public: a mixed methods study

Authors:

  Angela Ryan (a.v.ryan@bham.ac.uk)
  Sheila Greenfield (s.m.greenfield@bham.ac.uk)
  Sue Wilson (s.wilson@bham.ac.uk)

Version: 2 Date: 25 July 2006

Author's response to reviews:

Further to your email dated 18 July 2006, we would like to submit a revised manuscript. The changes are as detailed below:

We have removed 'Study Protocol' from the title page.

We have used sentence case in the title and all subheadings.

We have removed the telephone number of the corresponding author and the 'Keywords' section from the title page.

We have removed the word count from the 'Abstract'.

In the manuscript text, we have placed all references before the full stop or comma.

We have included details about funding in the 'Acknowledgements' section.

We have moved the details of ethical approval to the 'Methods' section.

The list of abbreviations is now called 'Abbreviations'.

In the 'Authors' contribution' section, we have indicated that all authors read and approved the final manuscript.

We have declared that we have no competing interests.

In the 'Acknowledgements' section, we have added the source of funding during the protocol development.

We have removed the dividing lines contained within the reference section.

We have also taken this opportunity to make minor changes to the text of the protocol, and we have corrected some references.

Please let me know if you require any further details of these amendments, or any further amendments.